Author’s response to reviews

Title: Validation of the German Revised Version of the Program in Palliative Care Education and Practice Questionnaire (PCEP-GR)

Authors:

Katharina Fetz (katharina.fetz@uni-wh.de)
Ursula Wenzel-Meyburg (wenzel-meyburg@med.uni-duesseldorf.de)
Christian Schulz-Quach (christian.schulz-quach@kcl.ac.uk)

Version: 3 Date: 04 Sep 2017

Author’s response to reviews:

Revision Letter 3

Validation of the PCEP-GR (Fetz, Wenzel-Meyburg, & Schulz, 2017)

Dear Dr. Pasini,

Thank you very much for yet another very useful feedback towards our manuscript. In reference to your feedback we have made the following modifications in our manuscript:

Email addresses of all authors were added to the title page, keywords were moved after the abstract and the statement of trial registration was removed. Abbreviations section was moved before the declarations section. The section Ethics and consent to participate was modified as suggested. A section consent for publication was added. A funding section was added. Figures and tables were removed from the manuscript and uploaded separately. Figure and table legends were added. The point-to-point-response was removed from the supplementary material. A clean version of the manuscript is submitted.

We hope that our manuscript is acceptable and suitable for publication now.

If there are still some minor changes necessary please give us a short feedback.

With kind regards,

On behalf of the author team

Katharina Fetz, MSc